Author's response to reviews

Title: Clinical significance of retinal emboli during diagnostic and therapeutic cardiac catheterization in patient with coronary artery disease

Authors:

javad kojuri (kojurij@yahoo.com)
morteza mehdizadeh (mehdizadeh@sums.ac.ir)
hamed rostami (rostam56@gmail.com)
danial shahidian (shahidd@yahoo.com)

Version: 6 Date: 14 January 2011

Author's response to reviews:

Dear Editors:

Thank you, I changed the format of table, ommit the legend of figure and provide functioning link to reference 18

sincerely Kojuri J